Author's response to reviews

Title: Prevalence of Complementary and Alternative Medicine use in Children with Chronic Health Conditions in Lagos, Nigeria

Authors:

Kazeem A Oshikoya (med_modhospital@yahoo.com)
Idowu O Senbanjo (senbanjo001@yahoo.com)
Olisamedua F Njokanma (faconi@yahoo.com)
Ayo Soipe (ayosoipeonline@yahoo.com)

Version: 5 Date: 11 December 2008

Author's response to reviews: see over
On behalf of my co-authors, I thank you for accepting in principle to publish our manuscript entitled “Prevalence of complementary and alternative medicine use in children with chronic health conditions in Lagos, Nigeria”. I am pleased to inform you the suggestions made by the second reviewer on pages 5 and 10 had been implemented. The manuscript has been copy edited at the University of Aberdeen (See additional document) and properly structured to conform with your instructions for authors. The background section of the abstract has been reviewed in line with current literature on CAM use in children.

I hope this revised manuscript will meet your requirements.

Yours sincere,

Dr Kazeem A. Oshikoya.